%================================================================
% LaTeX file with prefered layout for H1 paper drafts
% use: dvips -D600 file-name
%================================================================
\documentclass[12pt]{article}
\usepackage{epsfig}
\usepackage{amsmath}
\usepackage{hhline}
\usepackage{amssymb}
\usepackage{times}
\usepackage{cite}
\usepackage{multirow}
\renewcommand{\topfraction}{1.0}
\renewcommand{\bottomfraction}{1.0}
\renewcommand{\textfraction}{0.0}
\newlength{\dinwidth}
\newlength{\dinmargin}
\setlength{\dinwidth}{21.0cm}
\textheight23.5cm \textwidth16.0cm
\setlength{\dinmargin}{\dinwidth}
\setlength{\unitlength}{1mm}
\addtolength{\dinmargin}{-\textwidth}
\setlength{\dinmargin}{0.5\dinmargin}
\oddsidemargin -1.0in
\addtolength{\oddsidemargin}{\dinmargin}
\setlength{\evensidemargin}{\oddsidemargin}
\setlength{\marginparwidth}{0.9\dinmargin}
\marginparsep 8pt \marginparpush 5pt
\topmargin -42pt
\headheight 12pt
\headsep 30pt \footskip 24pt
\parskip 3mm plus 2mm minus 2mm
%===============================title page=============================
\begin{document}
\begin{thebibliography}{9}

\bibitem{sv00} H1 Collaboration, contributed paper 082 to EPS03, Aachen.

\bibitem{ISRold} H1 Collaboration, contributed paper 976 to ICHEP02, Amsterdam.

\bibitem{tomas} T. La\v{s}tovi\v{c}ka,
Eur.~Phys.~J. C {\bf 24} (2002) 529 [hep-ph/0203260].

\bibitem{kgbw} K. Golec-Biernat and M. W\"usthoff, Phys.~Rev. D {\bf 59}
(1999) 014017, [hep-ph/9807513].

\bibitem{allm} H. Abramowicz and A. Levy,
DESY-97-251, [hep-ph/9712415].

\bibitem{bpt} ZEUS Collaboration,
Phys.~Lett. B {\bf 487} (2000) 53.

\bibitem{qedc} H1 Collaboration, contributed paper 084 to EPS03, Aachen.

\bibitem{nmc} NMC Collaboration, Nucl.~Phys. {\bf  B483} (1997) 3.

\bibitem{f29697} H1 Collaboration, 
Eur.~Phys.~J. C {\bf 21} (2001) 33 [hep-ex/0012153]; \\
H1 Collaboration,
Phys.~Lett. B {\bf 520} (2001) 183 [hep-ex/0108035]. 

\end{thebibliography}

% The rest
\newcommand{\pom}{{I\!\!P}}
\newcommand{\reg}{{I\!\!R}}
\newcommand{\slowpi}{\pi_{\mathit{slow}}}
%\newcommand{\gevsq}{\mathrm{GeV}^2}
\newcommand{\fiidiii}{F_2^{D(3)}}
\newcommand{\fiidiiiarg}{\fiidiii\,(\beta,\,Q^2,\,x)}
\newcommand{\n}{1.19\pm 0.06 (stat.) \pm0.07 (syst.)}
\newcommand{\nz}{1.30\pm 0.08 (stat.)^{+0.08}_{-0.14} (syst.)}
\newcommand{\fiidiiiful}{F_2^{D(4)}\,(\beta,\,Q^2,\,x,\,t)}
\newcommand{\fiipom}{\tilde F_2^D}
\newcommand{\ALPHA}{1.10\pm0.03 (stat.) \pm0.04 (syst.)}
\newcommand{\ALPHAZ}{1.15\pm0.04 (stat.)^{+0.04}_{-0.07} (syst.)}
\newcommand{\fiipomarg}{\fiipom\,(\beta,\,Q^2)}
\newcommand{\pomflux}{f_{\pom / p}}
\newcommand{\nxpom}{1.19\pm 0.06 (stat.) \pm0.07 (syst.)}
\newcommand {\gapprox}
   {\raisebox{-0.7ex}{$\stackrel {\textstyle>}{\sim}$}}
\newcommand {\lapprox}
   {\raisebox{-0.7ex}{$\stackrel {\textstyle<}{\sim}$}}
\def\gsim{\,\lower.25ex\hbox{$\scriptstyle\sim$}\kern-1.30ex%
\raise 0.55ex\hbox{$\scriptstyle >$}\,}
\def\lsim{\,\lower.25ex\hbox{$\scriptstyle\sim$}\kern-1.30ex%
\raise 0.55ex\hbox{$\scriptstyle <$}\,}
\newcommand{\pomfluxarg}{f_{\pom / p}\,(x_\pom)}
\newcommand{\dsf}{\mbox{$F_2^{D(3)}$}}
\newcommand{\dsfva}{\mbox{$F_2^{D(3)}(\beta,Q^2,x_{I\!\!P})$}}
\newcommand{\dsfvb}{\mbox{$F_2^{D(3)}(\beta,Q^2,x)$}}
\newcommand{\dsfpom}{$F_2^{I\!\!P}$}
\newcommand{\gap}{\stackrel{>}{\sim}}
\newcommand{\lap}{\stackrel{<}{\sim}}
\newcommand{\fem}{$F_2^{em}$}
\newcommand{\tsnmp}{$\tilde{\sigma}_{NC}(e^{\mp})$}
\newcommand{\tsnm}{$\tilde{\sigma}_{NC}(e^-)$}
\newcommand{\tsnp}{$\tilde{\sigma}_{NC}(e^+)$}
\newcommand{\st}{$\star$}
\newcommand{\sst}{$\star \star$}
\newcommand{\ssst}{$\star \star \star$}
\newcommand{\sssst}{$\star \star \star \star$}
\newcommand{\tw}{\theta_W}
\newcommand{\sw}{\sin{\theta_W}}
\newcommand{\cw}{\cos{\theta_W}}
\newcommand{\sww}{\sin^2{\theta_W}}
\newcommand{\cww}{\cos^2{\theta_W}}
\newcommand{\trm}{m_{\perp}}
\newcommand{\trp}{p_{\perp}}
\newcommand{\trmm}{m_{\perp}^2}
\newcommand{\trpp}{p_{\perp}^2}
\newcommand{\alp}{\alpha_s}

\newcommand{\alps}{\alpha_s}
\newcommand{\sqrts}{$\sqrt{s}$}
\newcommand{\LO}{$O(\alpha_s^0)$}
\newcommand{\Oa}{$O(\alpha_s)$}
\newcommand{\Oaa}{$O(\alpha_s^2)$}
\newcommand{\PT}{p_{\perp}}
\newcommand{\JPSI}{J/\psi}
\newcommand{\sh}{\hat{s}}
%\newcommand{\th}{\hat{t}}
\newcommand{\uh}{\hat{u}}
\newcommand{\MP}{m_{J/\psi}}
%\newcommand{\PO}{\mbox{l}\!\mbox{P}}
\newcommand{\PO}{I\!\!P}
\newcommand{\xbj}{x}
\newcommand{\xpom}{x_{\PO}}
\newcommand{\ttbs}{\char'134}
\newcommand{\xpomlo}{3\times10^{-4}}
\newcommand{\xpomup}{0.05}
\newcommand{\dgr}{^\circ}
\newcommand{\pbarnt}{\,\mbox{{\rm pb$^{-1}$}}}
\newcommand{\gev}{\,\mbox{GeV}}
\newcommand{\WBoson}{\mbox{$W$}}
\newcommand{\fbarn}{\,\mbox{{\rm fb}}}
\newcommand{\fbarnt}{\,\mbox{{\rm fb$^{-1}$}}}
%
% Some useful tex commands
%
\newcommand{\qsq}{\ensuremath{Q^2} }
\newcommand{\gevsq}{\ensuremath{\mathrm{GeV}^2} }
\newcommand{\et}{\ensuremath{E_t^*} }
\newcommand{\rap}{\ensuremath{\eta^*} }
\newcommand{\gp}{\ensuremath{\gamma^*}p }
\newcommand{\dsiget}{\ensuremath{{\rm d}\sigma_{ep}/{\rm d}E_t^*} }
\newcommand{\dsigrap}{\ensuremath{{\rm d}\sigma_{ep}/{\rm d}\eta^*} }
% Journal macro
\def\Journal#1#2#3#4{{#1} {\bf #2} (#3) #4}
\def\NCA{\em Nuovo Cimento}
\def\NIM{\em Nucl. Instrum. Methods}
\def\NIMA{{\em Nucl. Instrum. Methods} {\bf A}}
\def\NPB{{\em Nucl. Phys.}   {\bf B}}
\def\PLB{{\em Phys. Lett.}   {\bf B}}
\def\PRL{\em Phys. Rev. Lett.}
\def\PRD{{\em Phys. Rev.}    {\bf D}}
\def\ZPC{{\em Z. Phys.}      {\bf C}}
\def\EJC{{\em Eur. Phys. J.} {\bf C}}
\def\CPC{\em Comp. Phys. Commun.}

\begin{titlepage}

\noindent
%%%Date:     03/31/2004 \\
%%%Version:  1.0        \\
%%%Editors:  Gerhard Brandt, Christian Veelken, Stefania Xella (junior editor: CD)\\
%%%Referees: Hinrich Meyer, David South \\
%%%%Comments by

%%%\vspace{2cm}

\vspace{1cm}
\begin{center}
{\large \bf H1prelim-04-042 }
\end{center}
\vspace{1cm}

\begin{center}
\begin{Large}

{\bf Measurement of the Inclusive DIS Cross Section at Low {\boldmath $Q^2$}
and High {\boldmath $x$} using Events with Initial State Radiation}

\vspace{2cm}

H1 Collaboration

\end{Large}
\end{center}

\vspace{2cm}


\begin{abstract}
The inclusive DIS cross section is measured at low $Q^2$ and relatively large
$x$ using events with initial state photon radiation from the 
incoming electron. In this analysis the radiated photon is not explicitly
detected. Instead its energy is inferred from a longitudinal momentum 
imbalance, such that the energy of the interacting electron and the event
kinematics can be reconstructed. The neutral current cross section
is thus measured
for $0.35 \leq Q^2 \leq 0.85 \ {\rm GeV^2}$
and $10^{-4} \ \lapprox \ x \ \lapprox \ 5 \cdot 10^{-3}$.
\end{abstract}

\vspace{1.5cm}

\begin{center}
Prepared for DIS 2004, Strbske Pleso, Slovakia
\end{center}

\end{titlepage}

%\newpage

\section*{Brief Analysis Details}

These results represent an extension to our previous measurements \cite{sv00}
of inclusive DIS cross sections at $Q^2 \ \lapprox \ 1 \ {\rm GeV^2}$,
obtained using a dedicated sample corresponding to around 
$600 \ {\rm nb^{-1}}$ for which the $ep$ interaction region was shifted
by $70 \ {\rm cm}$ in the outgoing proton direction. The shifted
vertex extends the detector
acceptance to smaller electron scattering angles and hence 
to lower $Q^2$ values. In the new analysis, an extension to larger $x$ values
at low $Q^2$ is achieved by making use of events with initial state photon 
radiation (ISR) from the incoming
electron. The photon radiation results in a reduced effective incoming
electron energy, such that the $ep$ centre of mass energy is reduced and
larger values of $x$ are accessed at fixed $Q^2$ than is the case for
non-radiative events. 

In contrast to earlier ISR analyses \cite{ISRold}, this analysis does not
require the observation of the radiated photon. Instead, its presence is
inferred from energy and longitudinal momentum conservation. 
Assuming that the photon is radiated collinear with the electron beam, 
the energy
$E_\gamma$ of the radiated photon is then given by 
$2 E_\gamma = 2 E_e^0 - (E-p_z)_{e^\prime} - (E-p_z)_h$, where 
$E_e^0 = 27.6 \ {\rm GeV}$ is the electron beam energy, 
$(E-p_z)_{e^\prime}$ is the measured difference between the energy and
the longitudinal momentum of the scattered electron and
$(E-p_z)_h$ is the same quantity for the full hadronic final state. The
reduced incoming electron beam energy $E_e$ is then given by 
$E_e = E_e^0 - E_\gamma$.
The reconstruction of $y$ and $Q^2$ proceeds using the 
`$\Sigma$ method', which relies on $\Sigma = (E - p_z)_h$. The reconstruction
of $x$ is then performed in a manner that is independent
of the electron beam energy $E_e^0$. The kinematics are resonstructed using
\begin{eqnarray*}
  y_\Sigma = \frac{\Sigma}{\Sigma + E_{e^\prime} (1 - \cos \theta_e)}
  \hspace*{1.5cm}
  Q^2_\Sigma = \frac{E_{e^\prime}^2 \sin^2 \theta_e}{1 - y_\Sigma}
  \hspace*{1.5cm}
  x_R = \frac{Q^2_\Sigma}{2 \Sigma E_p}
\end{eqnarray*}
where $\theta_e$ is the electron scattering angle. 
The remaining details of the analysis procedure closely
follow those reported in \cite{sv00}.

Control distributions corresponding to the measured ISR-dominated 
phase space are shown in figure~\ref{one}. 
There is good agreement between the observed uncorrected data distributions
and the expectation based on a DJANGO simulation of inclusive DIS together
with a PHOJET simulation of the photoproduction background. 

Figure~\ref{two} shows
the resulting measurements, expressed in the form of the reduced
cross section, 
\begin{eqnarray*}
  \sigma_r & = & \frac{Q^4 x}{2 \pi \alpha^2 Y_+} \ \
                 \frac{{\rm d} \sigma}{{\rm d} x {\rm d} Q^2} \\
           & = & F_2(x, Q^2) - \frac{y^2}{Y_+} \cdot F_L(x, Q^2) \ ,
\end{eqnarray*}
where $Y_\pm = 1 \pm (1 - y)^2$.
The data points have total uncertainties of around $10 \%$
and cover the region
$0.35 \leq Q^2 \leq 0.85 \ {\rm GeV^2}$
and $10^-4 \ \lapprox \ x \ \lapprox \ 5 \cdot 10^{-3}$. They are
compared with the extrapolated
predictions of a fit for $F_2$ based on a self-similar
structure to the proton (FRACTAL FIT $F_2$) \cite{tomas},
to which an $F_L$ term has also been added, based on a
dipole model \cite{kgbw}.They are also compared with the results from a 
parameterisation (ALLM97 $F_2$) \cite{allm} of inclusive
and exclusive DIS and photoproduction data. All predictions
are in good agreement with the data. 

The extension in
kinematic phase space achieved using the new method is 
illustrated in figure~\ref{three}, where the present data are
compared with the results from the non-radiative shifted
vertex analysis \cite{sv00}. Figure~\ref{four} shows the
overall status of DIS measurements in the low $Q^2$ region,
including in addition data from the ZEUS beampipe 
tracker \cite{bpt}, from a further H1 measurement at low $Q^2$ and
large $x$
using QED-Compton events \cite{qedc} and from NMC \cite{nmc}.
The new data are compatible with the previous measurements.

The new data allow improved extractions of the parameter
$\lambda$, which describes the growth of $F_2$ at low
$x$ according to $F_2 = c(Q^2) \cdot x^{- \lambda (Q^2)}$. 
Figure~\ref{five} shows the results for $\lambda$ obtained
by fitting the current data
at fixed $Q^2$ values, together with the previous
shifted vertex measurements \cite{sv00}. Also shown are
H1 results from larger $Q^2$ values \cite{f29697} and
ZEUS results \cite{bpt}. The new measurements
confirm the change in behaviour of $\lambda$ from a logarithmic
dependence on $Q^2$ at large $Q^2$ to a weaker dependence
compatible with reaching a constant consistent with the soft
pomeron intercept as $Q^2 \rightarrow 0$. The change takes
place on distance scales of the order of $0.3 \ {\rm fm}$
and can be interpreted as being related to a transition 
from partonic to hadronic degrees of freedom. 



\newpage

\begin{figure}[hhh]
\setlength{\unitlength}{1mm}
\begin{center}
\begin{picture}(150,150)(0,0)
\put(0,-20){\epsfig{file=H1prelim-04-042.fig1.eps, width=\textwidth}}
\end{picture}
\end{center}
\caption{Control distributions for the ISR-dominated event selection.
The uncorrected data distributions are compared with the sum of 
the DIS expectation (DJANGO Monte Carlo model) and the expected photoproduction
background (PHOJET Monte Carlo model). The comparisons are shown for the 
effective incoming 
electron energy $E_e$, the electron scattering angle $\theta_e$, 
the sum of the electron and
hadron $E - p_z$ and the reconstructed $y$, $Q^2$ and $x$ as used in the
measurement.}
\label{one}
\end{figure}

\newpage

\begin{figure}[hhh]
\setlength{\unitlength}{1mm}
\begin{center}
\begin{picture}(150,150)(0,0)
\put(0,-30){\epsfig{file=H1prelim-04-042.fig2.eps, width=\textwidth}}
\end{picture}
\end{center}
\caption{Measured reduced cross section from the shifted vertex ISR analysis,
compared with the extrapolated predictions of a `fractal' fit based on
self-similar proton structure, with the sum of the fractal fit result for
$F_2$ and a $F_L$ contribution from a dipole model and with the ALLM97
parameterisation.}
\label{two}
\end{figure}

\newpage

\begin{figure}[hhh]
\setlength{\unitlength}{1mm}
\begin{center}
\begin{picture}(150,150)(0,0)
\put(0,-30){\epsfig{file=H1prelim-04-042.fig3.eps, width=\textwidth}}
\end{picture}
\end{center}
\caption{Measured reduced cross section from both ISR and non-radiative
measurements using shifted vertex data.}
\label{three}
\end{figure}

\newpage

\begin{figure}[hhh]
\setlength{\unitlength}{1mm}
\begin{center}
\begin{picture}(150,150)(0,0)
\put(0,-30){\epsfig{file=H1prelim-04-042.fig4.eps, width=\textwidth}}
\end{picture}
\end{center}
\caption{Compilation of reduced cross section measurements 
with $Q^2 < 1 \ {\rm GeV^2}$ from H1, ZEUS and NMC.}
\label{four}
\end{figure}

\newpage

\begin{figure}[hhh]
\setlength{\unitlength}{1mm}
\begin{center}
\begin{picture}(150,150)(0,0)
\put(0,-30){\epsfig{file=H1prelim-04-042.fig5.eps, width=\textwidth}}
\end{picture}
\end{center}
\caption{Compilation of selected HERA results on the parameter $\lambda$, 
obtained from fits of the form $F_2 = c(Q^2) \cdot x^{- \lambda (Q^2)}$ to
low $x$ data.}
\label{five}
\end{figure}

\end{document}




